

Format for ANSWERING REVIEWERS



June 29, 2013

Dear Editor,

Title: Milligan-Morgan hemorrhoidectomy with anal cushion suspension and partial internal sphincter resection for circumferential mixed hemorrhoids

Author: Ming Lu, Guang-Ying Shi, Guo-Qiang Wang, Yan Wu, Yang Liu, Hao Wen

Name of Journal: *World Journal of Gastroenterology*

ESPS Manuscript NO: 3761

The manuscript has been improved according to the suggestions of reviewers:

1 Format has been updated

2 Revision has been made according to the suggestions of the reviewer

(1) References have been updated and supplemented

(2) Results section has been described in detail

(3) Article 72 cases belong to a clerical error, should be changed to 96 cases

(4) "A study of 96 cases" in Title of the article has been removed According to experts'

recommendations

(5) The chi-square value of each indicator and p values have been described in detail, as detailed in the table

(6) content in the Table 2 have been detailed description in the results section ,and removed the table 2

(7) Telephone have been supplement ,Fax none

3 References and typesetting were corrected

Thank you again for publishing our manuscript in the *World Journal of Gastroenterology*.

Sincerely yours,

Lu Ming, MD

Department of General Surgery,

First Affiliated Hospital

Xinjiang Medical University

E-mail: luming123luming@163.com